

Does Environmental Information Disclosure Reduce PM2.5 Emissions? Evidence from Chinese Prefecture-Level Cities

Teng Wang, Yani Wang, Weiwei Xiong

► To cite this version:

Teng Wang, Yani Wang, Weiwei Xiong. Does Environmental Information Disclosure Reduce PM2.5 Emissions? Evidence from Chinese Prefecture-Level Cities. Sustainability, 2024, 16 (22), pp.10125. 10.3390/su162210125 . hal-04802469

HAL Id: hal-04802469 https://hal.science/hal-04802469v1

Submitted on 25 Nov 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





Article Does Environmental Information Disclosure Reduce PM2.5 Emissions? Evidence from Chinese Prefecture-Level Cities

Teng Wang ^{1,2,*}, Yani Wang ² and Weiwei Xiong ³

- ¹ School of Economic Management and Law, Hubei Normal University, Huangshi 435002, China
- ² School of Economics and Management, China University of Geosciences, Wuhan 430074, China; wangyani1021@163.com
- ³ Laboratoire des Sciences du Climat et de l'Environnement (LSCE), Université Paris-Saclay, 91191 Gif-sur-Yvette, France; weiwei.xiong@lsce.ipsl.fr
- Correspondence: wangteng@cug.edu.cn

Abstract: As an important means of regulating pollution emissions, environmental regulation is crucial for reducing urban PM2.5. However, previous studies have mainly focused on the emission reduction effect of formal environmental regulations and neglected the role played by informal environmental regulations represented by environmental information disclosure. We employed a multiperiod difference-in-differences (DID) model to assess the effectiveness of EID policies in mitigating PM2.5 emissions and to investigate their abatement mechanism by focusing on green innovation and industrial structure. The findings indicate that the implementation of EID policies significantly reduces PM2.5 emissions. Mechanism tests reveal that EID promotes PM2.5 reductions by fostering green innovation and upgrading industrial structure. In addition, the impact of EID policy is more marked in resource-based cities and those located in interior regions. This study contributes to the reduction in urban haze emissions in China, offering empirical evidence and policy recommendations for the further implementation of environmental information disclosure.

Keywords: environmental information disclosure; PM2.5 emissions; prefecture-level cities; multiperiod DID model; mediating effect; heterogeneous effect

1. Introduction

China's economy has been in a period of rapid development since its reform and opening up. As a result of years of reckless economic growth, the environment has been damaged, and haze pollution has become a particularly worrying problem. According to the *National Ecological Quality Profile* promulgated by China's Ministry of Ecology and Environment, the national ambient air quality compliance rate in cities across the country was 46.6% in 2019, an increase of 25% from 21.6% in 2015. Data from the 2019 *China Ecological Environment Status Bulletin* show that the PM2.5 concentration in 337 cities at the prefecture and above levels across the country was 36 μ g/m³, unchanged from the previous year, with 46.6% of all cities meeting the environmental air quality standard [1]. In 2019, these 337 cities together had 452 days of severe pollution, 183 days fewer than in 2018, and 1666 days of heavy pollution, 88 days more than in 2018. On 78.8% of the days with severe or worse pollution, PM2.5 was the main pollutant [2]. These data indicate that the sources of PM2.5 pollution remain a top target in China's air pollution management efforts.

In response, the Chinese government prioritized this issue by issuing a series of measures to reduce PM2.5 concentrations, such as the *Air Pollution Prevention and Control Law*, the *Air Pollution Prevention and Control Action Plan*, and the *Ambient Air Quality Standards*. One key measure, the 2013 *Air Pollution Prevention and Control Action Plan*, is considered one of the most ambitious environmental policies in China's history. It has significantly curbed haze pollution but at a considerable cost. Taken together, these policies and regulations have helped to improve China's air quality, but progress remains insufficient.



Citation: Wang, T.; Wang, Y.; Xiong, W. Does Environmental Information Disclosure Reduce PM2.5 Emissions? Evidence from Chinese Prefecture-Level Cities. *Sustainability* **2024**, *16*, 10125. https://doi.org/10.3390/ su162210125

Academic Editor: Luca Stabile

Received: 26 June 2024 Revised: 14 November 2024 Accepted: 19 November 2024 Published: 20 November 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The sustained effectiveness of the air pollution control measures led by the Chinese government is hampered by information asymmetry between local governments and polluting enterprises, leading to frequent instances of enterprises evading environmental supervision [3]. The effectiveness of environmental governance can be determined by a robust EID system that facilitates joint governance by the government, corporations, and the public. EID enhances public awareness of environmental protection, motivates government oversight, establishes internal and external mechanisms for environmental supervision, resolves information asymmetry in pollution control processes, and improves regional pollution control performances. According to the theory of signal transmission, credible, comprehensive, transparent, and high-quality environmental information is needed for effective governmental pollution prevention and control [4].

Present studies on Chinese environmental protection policies and their effects on PM2.5 reductions tend to focus on the implementation of governmental environmental regulations, such as establishing regulatory frameworks [5], environmental protection supervision [6], and environmental justice [7,8], and on the level of market regulation, like environmental taxes [9], subsidies [10], and emissions trading [11], for pollution prevention and control. However, less research has examined the impact of pollution disclosure on reducing PM2.5 emissions in urban areas. Among the limited empirical studies exploring pollution governance through environmental information disclosure, the Pollution Information Transparency Index is commonly employed to measure the level of urban environmental transparency [12,13]. It should be noted that such evaluation indexes, derived from comprehensive index system calculations, often face issues of subjectivity, limited comprehensiveness, and timeliness [14]. Additionally, most cities that have been selected for evaluation are national key cities for environmental protection, leading to potential selection bias and skewed evaluation results [15].

The Chinese government is gradually establishing and improving its environmental information disclosure regime. In 2007, the State Environmental Protection Administration promulgated the *Environmental Information Disclosure Measures (for Trial Implementation)*, which came into effect on 1 May 2008. This marked the initial construction of China's environmental information disclosure system. By collecting panel data on 256 prefecture-level cities from 2005 to 2021, we constructed a quasi-natural experiment using the multiperiod DID method to investigate the EID effect on PM2.5 governance. We found that (1) EID can significantly reduce PM2.5 emissions; (2) EID indirectly contributes to the reduction in PM2.5 by increasing green innovation and upgrading industrial structures; and (3) the impact of EID on PM2.5 is more significant in resource-based and interior cities.

The key contributions of this study are in the following areas: First, existing research on PM2.5 pollution control and reduction predominantly concentrate on commandand-control regulations or market-based incentives, with few studies addressing publicparticipation-driven environmental regulation. This study centers on the Public Disclosure of Environmental Information (Trial), China's first official regulation on environmental information disclosure, which aims to promote public participation in the environmental protection. Doing so further expands the scope of research on China's environmental protection policies.

Second, from a methodological perspective, this study adopts a city–year sample selection and employs a DID model to assess the influence of EID policies on urban PM2.5 emissions. This method categorizes prefecture-level cities into the treatment or control group based on whether they have implemented EID policies, allowing for both horizontal comparisons between cities implementing the policy and those that did not, as well as vertical comparisons of cities before and after policy implementation. The DID approach mitigates the effect of external factors on PM2.5 emissions before and after policy implementation and effectively addresses the endogeneity issue between EID and PM2.5 emissions. This method enables a more precise identification of the net influence of EID policies on urban PM2.5 emissions.

Finally, this study highlights the significance of green innovation and industrial structure in shaping the linkage between EID and urban PM2.5 emissions. By organizing theoretical insights, we construct a conceptual framework that links EID, green innovation, industrial structure, and urban PM2.5 emissions. This framework offers a new perspective for understanding how EID influences urban PM2.5 emissions.

The rest of the study is structured in the following way: Section 2 reviews the policy background of EID; Section 3 reviews the relevant literature and presents the hypotheses; Section 4 outlines the model, data, and variables; Section 5 presents the results of the empirical analysis; and Section 6 discusses the conclusions and provides policy recommendations.

2. Policy Background

The high-energy-consumption, low-efficiency development pattern has promoted the rapid development of China's economy over the past decades. However, it has also led to serious environmental pollution problems, particularly air pollution, which have raised social concerns. High concentrations of PM2.5 in the air can severely degrade air quality and endanger human health [16]. In response to haze pollution, the Chinese government has adopted several environmental regulatory measures, resulting in various outcomes. EID is increasingly valued by governments and enterprises as a new means of environmental regulation in the context of economic development and improving living standards to further deepen environmental governance and promote economic transformation and transparency with regard to the environmental situation. On 8 February 2007, the Chinese government launched the Environmental Information Disclosure Measures (Trial), which mandates both government and enterprise environmental information disclosure. These measures aim to promote the construction of an EID system.

EID is widely practiced globally. Historically, EID first emerged in the form of legislation, stipulating the items to be disclosed by law and ensuring citizens' right to know [17]. In the US, for instance, the 1986 Environmental Protection and Consumer Protection Act (EPCRA) requires its manufacturers to disclose annually to the public the amount of toxic chemicals they released into the environment. This disclosure system is administered by the US Environmental Protection Agency (EPA) and has been attributed to a 46% reduction in toxic emissions over an 11-year period [18]. Practice has proven that the EID system meets governmental needs for environmental regulation and guarantees citizens' right to environmental information. It helps maintain social justice, protect and promote rational use of the environment and resources, coordinate the relationship between humans and nature, and, ultimately, achieve sustainable development.

As for specific measures of EID, the "Measures" made the following provisions: in terms of government information disclosure, environmental protection departments should take the initiative to disclose environmental protection laws and regulations, environmental quality conditions, the allocation of total emissions of major pollutants and their implementation, the issuance of emission permits, the quantitative assessment results of comprehensive urban environmental improvement, and pollutant emissions that exceed the standard list of polluting enterprises. In terms of corporate information disclosure, it implements a combination of voluntary and mandatory disclosure. It encourages enterprises to voluntarily disclose the total annual consumption of resources; the type, quantity, concentration, and destination of pollutants emitted by enterprises; the treatment and disposal of their production waste; and the recycling and recovery of their waste. Specific enterprises should also disclose the names of main pollutants, emission methods, and emission concentrations and totals, as well as instances where standards and total limits have been exceeded. In addition to the disclosure of content, it also includes manners and procedures to ensure the implementation of the EID system. This policy marked the initial establishment of China's EID system. It promotes China's transition from the traditional government-led system to a more transparent system of environmental governance.

Since the enactment of the EID policy in 2007, the quantifiable evaluation of its implementation has been a key issue in the further development of EID practice in China. Since 2008, environmental information on 113 key cities related to environmental protection has been published annually, evaluating the following eight items: "information on daily exceedance of standards and violation records of pollution sources", "information on centralized remediation of pollution sources", "information on clean production audits", "information on the overall evaluation of the environmental behavior of enterprises", "information on letters, visits, and complaints from the public on environmental issues or pollution by enterprises and the results of their treatment as verified by investigation", "information on the acceptance of environmental impact assessment documents of construction projects and the results of environmental protection acceptance of completed construction projects", "information on sewage charges", and "information on public disclosure upon application".

3. Literature Review and Theoretical Hypotheses

3.1. Literature Review

The literature closely related to our study focuses primarily on the following two key areas: (1) relationships between environmental regulation and pollution and (2) environmental information disclosure.

3.1.1. Environmental Regulations and Pollution Emissions

Environmental regulation can be categorized into the following two types: formal and informal regulation [19]. Formal environmental regulation involves government use of administrative measures to guide enterprise production and operational practices, aiming to achieve both sustainable development and environmental protection [20]. In contrast, informal environmental regulation refers to constraints imposed by stakeholders through public opinion and emissions monitoring, aiming to motivate enterprises to reduce pollution emissions [21].

The linkage between formal environmental regulation and environmental pollution is a prominent issue in academia, but there is still no consensus among researchers about their relationship. One viewpoint argues that formal environmental regulation facilitates pollution reduction. Properly designed environmental regulations compel high-energy-consuming, high-pollution enterprises to invest in technological innovation and adopt advanced pollution control technologies, ultimately reducing emissions [22,23]. The second view posits that formal environmental regulation may increase pollutant emissions, creating a "green paradox" [24]. Strictly enforced regulations raise the compliance costs for high-pollution, high-emission firms, crowding out funds for technological innovation. In pursuit of profit maximization, firms may even increase emissions by expanding production to maintain profitability [25]. A third perspective suggests that the relevance between formal environmental regulation and pollution emissions is not linear. Wang and Zhang (2022) [26] found that formal environmental regulation has a reverse U-shaped link with urban CO₂ emissions, as follows: initially, regulation increases emissions, but after the inflection point, regulation reduces emissions.

The effect of informal environmental regulation on pollution reduction has received increasing attention from researchers in recent years. Most scholars agree that informal environmental regulation is crucial for reducing pollution [27–29]. Mechanistically, Shen et al. (2023) [30] found that informal environmental regulation reduces regional CO₂ mainly through upgrading the industrial structure and substituting renewable energy. Cui and Cao (2024) [31] identified three channels—media environmental concern, corporate green innovation, and government environmental governance—through which informal environmental regulation reduces urban PM2.5 pollution. Liu et al. (2023) [32] noted that public concern, as an informal form of environmental regulation, reduces PM2.5 emissions by enhancing pollution detection. Furthermore, some researchers suggest that the link between informal environmental regulation and pollution is not simply linear but follows a reversed U-shaped curve [33].

3.1.2. Environmental Information Disclosure

Environmental pollution control requires collaboration among multiple stakeholders, and an EID policy serves as a governance tool that engages stakeholders, such as the government, corporations, and citizens. Existing research on the EID policy concentrates mainly on the following two dimensions: determinants and consequences.

Environmental information disclosure can be influenced by a range of factors. Lv et al. (2024) [34] show that establishing regional environmental courts can improve corporate environmental disclosure. New media monitoring and environmental information uncertainty were found to encourage corporate information disclosure by Zhang and Xiang (2024) [35]. Chen et al. (2024) [36] empirically confirm that investor attention dampens EID behavior in high-tech firms in China. Zhang et al. (2024) [37] argue that online media pressure and a robust internal control system can enhance corporate environmental information disclosure. Big data technology can improve the quality of corporate EID and tackle the challenge of misinformation according to Zhang et al. (2024) [38]. Wei et al. (2024) [39] found that management ownership is positively correlated with environmental disclosure.

The implementation of environmental information disclosure has varying effects at different levels. At the regional level, Yu and Jin (2022) [40] found that environmental information disclosure significantly enhances public environmental awareness in urban areas, with this effect increasing as EID expands. Pan et al. (2023) [41] argued that the enforcement of the EID policy contributed to the inflow of FDI into China's pilot cities, creating a siphoning effect. Wang et al. (2024) [42] found that the EID policy notably increased carbon total factor productivity in Chinese pilot cities, whereas Guo and Xu (2024) [43] demonstrated that the EID policy fosters urban economic growth by attracting talent capital. At the firm-level, Guo et al. (2023) [44] found that EID significantly reduces the financialization of nonfinancial firms by facilitating R&D investment, whereas Feng et al. (2024) [45] demonstrated that EID stimulates green innovation by reducing information asymmetry and alleviating financial constraints. Tan and Liu (2024) [46] argued that EID promotes corporate environmental performance, though this effect is weakened by analysts and media reports.

Various types of environmental regulations influence pollution emissions through different mechanisms and directions. In contrast to command-and-control or market-based environmental regulations, informal environmental regulations, such as environmental information disclosure policies, which encourage public participation, are gaining increasing importance. This study examines the impacts and mechanisms of EID on PM2.5 emissions, contributing to the growing literature on this topic.

3.2. Theoretical Hypothesis

3.2.1. Direct Effect Analysis

In 2008, the Environmental Information Disclosure Measures (Trial) came into effect in China. It required both the government and firms to disclose environmental information. At the government level, it is necessary to disclose to the public the environmental quality, allocation, and implementation of total emissions targets of major pollutants, issuance of emission licenses, and quantitative evaluation results of comprehensive urban environmental improvement in cities. At the firm level, enterprises are encouraged to voluntarily disclose environmental information, including the total annual consumption of resources and the type, amount, concentration, and destination of pollutants emitted. The procedures for disclosure also made provisions for using the government website, bulletins, press conferences, newspapers, radio, television, and other means to facilitate timely public access to environmental information. Thus, the environment can be improved and PM2.5 can be reduced through the following three aspects: government, enterprises, and the public.

First, EID imposes requirements on government actions. Under the governmental target responsibility assessment system, the government is guided by environmental management goals and administrative responsibilities, forming an incentive and constraint mechanism [47]. To complete the performance assessment, the government will take en-

vironmental problems seriously, thus promoting the reduction in PM2.5 concentrations. Second, enterprises' pollution information is much more transparent under the EID system. On the one hand, as the main emitters, enterprises will face increased production costs and reduced profits due to sewage charges. On the other hand, companies with excessive emissions will be listed as serious polluters, negatively impacting their reputation, leading to decreased investor confidence and reduced consumer demand for their products [48]. On the basis of stakeholder theory, enterprises must meet the requirements of different stakeholders, including the government, the public, investors, and environmental organizations. EID enables enterprises to address these concerns, thereby enhancing their reputation and market value [49]. To avoid negative consequences, companies take steps to reduce pollutants, thereby effectively controlling PM2.5 emissions. As EID progresses, the transparency of environmental information continues to improve. This increased transparency facilitates public access to environmental information, fostering societal oversight that holds both the government and enterprises accountable [50], ultimately improving environmental quality and reducing PM2.5 emissions. Therefore, we propose the following Hypothesis 1:

H1. EID helps directly reduce PM2.5 emissions.

3.2.2. Impact Mechanism Analysis

In general, EID promotes green innovation (GI) [51] and industrial structure (IS) upgrading [52], both of which have been identified as critical factors influencing pollution emissions [53,54]. Thus, this article primarily explores the mechanisms through which EID affects PM2.5 emissions across cities, focusing on the roles of green innovation and industrial structure.

(1) Green innovation effect

EID can stimulate innovation in various ways. It can reduce the cost of innovation through government incentives, lowering the threshold for firms to innovate and providing a better environment for innovation. It can also increase social media attention, enhancing social reputation [55] and attracting capital market stakeholders to push enterprises toward green innovation. A higher level of EID can reduce information asymmetry, lower investment risks, promote corporate financing, and provide more financial support for green innovation [56]. Environmentally sensitive consumer groups are also attracted to the high-quality environmental information revealed by firms [57]. These consumers tend to prefer environmentally friendly products, prompting companies to optimize production lines, improve production processes, develop more environmentally friendly products, and achieve more green innovations [58].

Green innovation plays a crucial role in achieving green development goals, encompassing both green and innovative attributes. The primary objective of green innovation must be environmental protection [59]. At the macro level, green innovation promotes the optimization and upgrading of energy structures, thereby reducing air pollutant emissions and PM2.5 concentrations [60]. At the firm level, green technological innovation encompasses advancements in production processes and governance [61]. The former focuses on green upgrades in production processes to maximize resource efficiency and reduce total pollutants, whereas the latter enhances pollution control technologies and treatment facilities to improve regional air quality and lower PM2.5 levels. Therefore, we propose the following Hypothesis 2:

H2. EID drives green innovation and, thus, reduces PM2.5.

(2) Industrial structure effect

The EID policies reduce the share of heavy industry in a city's GDP, facilitating the optimization and upgrading of urban industries. First, as the EID policy is carried out, pilot cities release detailed emission reduction methods and targets. They optimize industrial layouts by promoting pollution reduction, accelerating the elimination of outdated

production capacities in heavily polluting sectors, like electricity, steel, petrochemicals, and mining, and closing, merging, or upgrading the technology of polluting enterprises for energy efficiency and environmental protection [62]. Second, the policy raises environmental access thresholds, making the approval of new or expanded high-emission enterprises, such as steel mills, thermal power plants, and cement plants, more stringent, which further expedite the removal of outdated production capacities in the region [63]. Finally, the policy increases local government support for cleaner production and provides incentives for enterprises to develop environmentally friendly industries, thus promoting the orderly optimization and upgrading industrial structure by increasing the share of cleaner industries in the region [64].

Given that different industries vary in their levels of resource consumption and environmental damage, industrial upgrading reduces PM2.5 emissions by lowering pollutant emissions. Although industrial restructuring appears to involve changes in the proportion of each industry, it fundamentally reflects shifts in energy utilization and efficiency across industries. The secondary industry is dominated by energy-intensive heavy industries characterized by high resource consumption, low technological advancement, and inefficient development. As the proportion of the secondary industry in a city grows, resource consumption rises significantly, resulting in higher urban pollutant emissions. In contrast, the tertiary industry, which is dominated by the production and service sectors, is less reliant on fossil resources. An increase in the tertiary sector's share of a city's GDP reduces fossil energy demand and lowers pollutant emissions. Therefore, we propose the following Hypothesis 3:

H3. *EID drives industrial structure upgrading to reduce PM2.5.*

Based on the above theoretical analysis, the mechanism of EID on PM2.5 is shown in Figure 1.



Figure 1. Influencing mechanism of EID on PM2.5 reduction.

4. Methodology and Dataset

4.1. Methodology

4.1.1. DID Method

The difference-in-differences (DID) method is a commonly used method of causal inference that is primarily used to evaluate the influence of an event or policy [65]. The basic principle is to infer the causal effect of an intervention by comparison of changes in the experimental and control groups before and after the intervention. In the first step, a control group with a trend similar to that of the experimental group before the intervention is selected to ensure that the trends of the two groups before the intervention are parallel.

In the second step, the causal effect of the intervention was calculated by comparing the changes in the experimental and control groups before and after the intervention.

In 2008, the Public Environmental Research Center and the Natural Resources Defense Council jointly published the Pollutant Information Disclosure Index for 113 cities in China. This number expanded to 120 cities in 2013. To examine the effect of EID on PM2.5 in 256 Chinese prefecture-level cities, a quasi-natural experiment was constructed. The 116 cities (excluding the four municipalities directly under the Central Government) served as the experimental group, and the remaining 140 cities served as the control group. Since the time points of the policies were inconsistent, the net effect of EID on PM2.5 was assessed using a multiperiod DID approach. The model was set up as follows:

$$PM2.5_{it} = \alpha_0 + \alpha_1 did_{it} + \sum \alpha_j Controls_{jit} + \mu_i + \gamma_t + \varepsilon_{it}$$
(1)

In the above equation, *i* and *t* denote city and year, respectively. $PM2.5_{it}$ denotes the PM2.5 concentration, and did_{it} indicates whether the city is included in the assessment of EID, taking 1 for the experimental group and 0 for the control group. The coefficient α_1 represents the net effect of EID on PM2.5 emissions, which is the focus of this study. *Controls* represent a set of control variables affecting PM2.5. α_j denotes the coefficient of the control variable. μ_i and γ_t represent city- and time-fixed effects, respectively. ε_{it} is the random disturbance term.

4.1.2. Intermediary Effect Model

The regression results above confirm that EID can significantly reduce PM2.5, but what is the path of this influence? This study investigates the impact mechanism by constructing a two-step mediating effects model, with the first step being Equation (1) and the second step being the equation shown below:

$$M_{it} = \lambda_0 + \lambda_1 did_{it} + \sum \alpha_i Controls_{jit} + \mu_i + \gamma_t + \varepsilon_{it}$$
⁽²⁾

In the above equation, *M* is the mediating variable. The influencing mechanism was tested in terms of both the level of green innovation and industrial structure.

4.2. Data and Variables

The dataset used in this study consisted of a panel of prefecture-level cities in China, covering the years 2005–2021. The data are taken from the *Chinese Urban Statistical Yearbook* 2022. It is important to note that Beijing, Shanghai, Tianjin, and Chongqing are provincial-level administrative units with significantly higher political statuses than typical prefecture-level cities; thus, these four cities were excluded from the sample. Additionally, some cities were dropped due to lack of data. As a result, the final sample consists of 256 cities.

4.2.1. Dependent Variable

As the explained variable, Chinese officials only started ground-based monitoring of PM2.5 in some key cities in 2013, which means the data are unable to meet the needs of our study. In addition, the data obtained from ground-based monitoring are point-source data, which are difficult to use to measure the whole picture of PM2.5 emissions in the whole city [66,67]. Fortunately, Chinese PM2.5 data could be obtained from the Dalhousie Atmospheric Composition Analysis Group (Canada) and are based on global satellite monitoring of PM2.5 pollution [68]. The raster data of annual average concentrations were processed and measured with prefecture-level cities in China as the basic units.

4.2.2. Independent Variable

Since the element of "information" is hard to quantify and has its own endogeneity problem, this study chooses whether prefecture-level cities are included in the evaluation of EID (did) as the independent variable. Out of the 116 cities in the experimental group, 109 began to disclose information in 2008, whereas the remaining 7 began to disclose

information in 2013. If environmental information is disclosed in the current year, the value is 1; otherwise, the value is 0.

4.2.3. Control Variables

(1) Fixed-asset investment (FAX). FAX influences multiple aspects of national economic growth and serves as a key tool for adjusting the economic structure and optimizing productivity distribution [69]. Over the long term, fixed-asset investment lays a robust material and technological foundation for economic development, directly impacting PM2.5 emissions by phasing out outdated production methods and enhancing the production structure. At present, there are two main methods of measuring fixed asset investment: fixed asset investment per capita and fixed asset investment as a proportion of GDP. Consideration of the latter measurement method can, to a certain extent, reflect the pattern and strategy of local economic development. Changes in this indicator can reflect adjustments in the economic structure and changes in the direction of development, which are crucial to sustainable economic development. Therefore, the ratio of urban fixed-asset investment to GDP (Unit: %) is used to measure the level of fixed-asset investment [70].

(2) Foreign direct investment (FDI). Currently, scholars have not reached a consensus on the influence of FDI on PM2.5 emissions. The pollution haven hypothesis posits that developed countries use FDI to relocate polluting production activities to developing countries, leading the latter to bear the environmental pollution consequences [71]. In contrast, the pollution halo hypothesis argues that FDI introduces advanced technology and valuable production management expertise to the host country, thereby reducing its PM2.5 emissions [72]. There are two indicators of FDI in the City Statistical Yearbook, namely, number of projects for contracted foreign direct investment and amount of foreign capital actually utilized. The latter can more truly reflect the amount of foreign direct investment absorbed by the city. This study uses the proportion of amount of foreign capital actually utilized to GDP (Unit: %) as the indicator of the FDI level [73].

(3) Urbanization (Urb). Urb is a key factor influencing PM2.5 emissions. On the one hand, as urbanization progresses, a large portion of the rural population migrates to cities, increasing the demand for resources and energy that, in turn, exacerbates PM2.5 emissions in urban zones [74]. On the other hand, the expansion of urban areas attracts skilled professionals, enterprises, and research institutions, leading to the dissemination of knowledge, information, and technology, which promote technological advancement and help mitigate urban haze pollution [6]. There are three main methods for measuring urbanization: the urban population ratio method, the built-up area ratio method, and the construction of comprehensive indicators method. Considering that people are the core of urbanization, this article adopts population urbanization, which refers to the proportion of urban population to total population (Unit: %), to measure the level of urbanization [75].

(4) Fiscal Decentralization (FD). FD is an additional factor influencing PM2.5 emissions. Under the perspective of fiscal decentralization, local governments have an obvious information advantage and can formulate haze management policies that are in line with the economic development of their jurisdictions, so as to internalize the costs and benefits of management and improve the efficiency of PM2.5 emission reduction. There are two main measurement methods in existing studies: a revenue and expenditure perspective, which uses revenue decentralization and expenditure decentralization indicators to measure the degree of fiscal decentralization, and a self-sufficiency perspective, which measures the degree of fiscal decentralization by using the ratio of local general public budget revenues to local general public budget expenditures. The fiscal self-sufficiency perspective can fully capture the financial autonomy of cities in reducing PM2.5 emissions, because this paper finally chooses this method to measure the degree of fiscal decentralization (Unit: %) of cities [76].

(5) Economic development (ED). The level of economic development is one of the most important causes of urban PM2.5, but scholars have not yet reached a consensus on the relationship between the two. On the one hand, the higher the level of economic

development of a city, the more importance it attaches to the protection of the ecological environment, and thus it will take a variety of measures to reduce urban PM2.5 emissions [77]. On the other hand, economic growth will consume a large amount of fossil energy, which will emit more PM2.5 [78]. GDP and per capita GDP (PGDP) are often used to measure the level of economic development of a region; GDP is the total amount of economic activity in a region, while PGDP is treated per capita through the GDP total, which better reflects the average level of the region's economy and the quality of life of its people. We borrow from Jiang et al. (2022) [79] and use PGDP (Unit: CNY 10,000 / person) to measure the level of economic development.

4.2.4. Intermediate Variables

(1) Green innovation. Green innovation was measured by two types of indicators: green patent applications or green patent grants. Considering that green patents took a long time to go through from application to grant, the lag phenomenon was more obvious. This study used the number of green patent applications (Unit: 10,000 items) to measure green innovation [51]. Data were sourced from the patent database of the State Intellectual Property Office. Green patents were identified using the "Green Patent List", released by the World Intellectual Property Organization, in 2010, and the green patents of each city are aggregated from micro-level data of the listed companies.

(2) Industrial structure. The industrial structure serves as an important intermediary factor in the linkage between environmental information disclosure and urban PM2.5 emissions. Industries are the main driving force of and important support for the economic development of cities. Among the three industries, the secondary industry includes the chemical, electric power, and iron and steel industries, which are key sectors for fossil energy consumption, and the pollutants emitted during their production processes are the main cause of PM2.5 emissions. Following the measurement approach of Feng et al. (2021) [47], industrial structure was quantified by the secondary industry value-added share of GDP (Unit: %).

5. Results

5.1. Baseline Regression Analysis

5.1.1. Descriptive Statistics

Table 1 presents the descriptive statistics of the main variables. The range of urban PM2.5 emissions spans from 11.637 μ g/m³ to 108.955 μ g/m³, indicating significant disparities in emission levels across different cities. In addition, some cities face severe PM2.5 emission issues that require urgent attention. Additionally, the average PM2.5 emission level in these cities is 43.827 μ g/m³, significantly exceeding the national concentration limit of 15 μ g/m³ and the secondary limit of 35 μ g/m³, as per the ambient air quality standards. This indicates that China's prefecture-level cities are struggling to control PM2.5 emissions, with substantial room for improvement.

Variables	Ν	Mean	SD	Min	Max
PM2.5	4352	43.827	15.603	11.637	108.955
did	4352	0.320	0.467	0	1
FAX	4352	0.754	0.332	0.090	2.860
FDI	4352	0.018	0.020	0	0.229
Urb	4352	0.514	0.164	0.120	1
FD	4352	0.465	0.223	0.0260	1.256
ED	4352	10.418	0.751	4.595	13.056

Table 1. Cont.

Variables	Ν	Mean	SD	Min	Max
GI	4352	0.023	0.074	0	1.375
IS	4352	0.469	0.107	0.107	0.859

Note: The minimum value of the variable FDI is not 0; however, it is displayed as 0 because it cannot be represented with 3 decimal places in the table.

5.1.2. Baseline Results

Table 2 presents the results of the baseline regression. The Hausman test results indicate a fixed-effects model is appropriate for estimating the panel data. Both column (1) and column (2) account for year- and city-fixed effects. Column (1) does not control for characteristic city variables, and the coefficient of *did* is -1.245, significant at the 1% level, indicating that EID contributes to the reduction in PM2.5. Column (2) includes city characteristic variables, and the estimated results show that the significant impact remains unchanged. In column (2), the coefficient of the *did* is -1.586, also significant at the 1% level. This indicates that if a city discloses environmental information, PM2.5 emissions will decrease by 1.586 µg/m³. This finding confirms the hypothesis of the previous study.

Table 2. Results of benchmark regression analysis.

Variable	Explained Variable: PM2.5		
variable	(1)	(2)	
did	-1.245 *** (0.303)	-1.586 *** (0.308)	
FAX		-1.517 *** (0.356)	
FDI		-16.717 *** (5.815)	
Urb		-4.254 *** (1.253)	
FD		-5.824 *** (1.046)	
ED		-0.483 (0.378)	
City FE	Yes	Yes	
Year FE	Yes	Yes	
Constant	44.226 *** (0.115)	55.711 *** (3.684)	
N	4352	4352	
R ²	0.936	0.938	

Note: *** indicate significance levels of 1%. The values in parentheses are the robust standard error of regression coefficient.

Pollution management requires the joint efforts of different stakeholders, and EID is a management tool that involves the participation of multiple stakeholders, including government, enterprises, and citizens. EID has a positive impact on PM2.5 management for several reasons. First, with the disclosure of pollution information, enterprises, as the main emitters, are monitored by the government and the public. To avoid severe administrative penalties and pressure from public opinion, enterprises are motivated to invest in reducing pollution emissions by improving technology. Second, information gaps and uncertainties are core problems of pollution management, and disclosing environmental information significantly reduces the concealment and nonreporting of pollution. Third, increased disclosure of environmental information satisfies the public's right to know, and public participation has been shown to help improve environmental efficiency.

5.1.3. Parallel Trend Test

The parallel trend test is a precondition for using the difference-in-differences method. This presupposes that the experimental and control groups satisfy the parallel development assumption before the policy is implemented. This means that we need to examine how the experimental and control groups differ before the policy is implemented. Only when they are similar enough can the reliability of the experimental results be guaranteed. Changes in trends in the three years before and the eight years after EID implementation are examined in this study. The results of the parallel trend test are shown in Figure 2. Prior to the EID implementation, the experimental and control groups exhibited consistent trends with no significant differences. Following the policy's enactment, the experimental group experienced a significant reduction in PM2.5 concentration compared to the control group. This confirms that the sample passed the parallel trend test, indicating the EID policy as a crucial factor in reducing PM2.5 emissions.



Figure 2. Parallel trend test results. Note: As a result of the two policy shocks in 2008 and 2013, there are limited data for periods 8 to 3 before policy implementation and periods 10 to 13 after policy implementation. To facilitate the analysis, the data from periods 8 to 4, prior to the policy's implementation, were combined with the 3rd period, whereas the data from periods 10 to 13, after the policy's implementation, were grouped with the 9th period. Additionally, the third period, before the policy's implementation, was used as the baseline period to derive the results presented above.

5.2. Analysis of Impact Mechanism

Building on the theoretical analysis and the model's construction in the previous section, this section explores the policy's mechanism through the following two key dimensions: green innovation and industrial structure.

5.2.1. Mechanistic Test for Green Innovation

Green innovation was incorporated as a mediating variable in the regression model, and the results are presented in Table 3. The results of the benchmark regression in the second column are consistent with the results in the last column of Table 2. Building on this, green innovation is included as a mediating variable. In addition, the results are shown in the third column of Table 3. The regression coefficient between EID and GI is 0.037, passing the 1% significance level, indicating that the enforcement of EID policy strongly promotes urban green innovation. This demonstrates that green innovation partially mediates the effect of EID on PM2.5. These regression results conclude that EID impacts PM2.5 through the mediation of green innovation, confirming Hypothesis 2.

EID fosters green innovation by enhancing cities' competitiveness in green development and stimulating green innovation vitality. Public disclosure of environmental information exerts pressure on both local governments and companies. Governments, aiming to preserve their reputation, adopt targeted policies to address environmental issues. Similarly, companies must safeguard their social image and business interests, prompting them to pursue green technological innovation for sustainable development.

Green innovation has a crucial role to play in environmental protection. Du and Wang (2022) [49] provide both theoretical and empirical evidence for the role of green innovation in mitigating haze pollution. In the production stage, green innovation enhances energy efficiency, facilitating the shift from fossil fuels to clean energy, thereby reducing emissions. In the treatment stage, green innovation supports the management and storage of emissions.

Overall, EID has bolstered firms' innovation performance and contributed to reducing PM2.5 emissions.

Variable		Intermediate Variables		
variable	P1v12.5	GI	IS	
did	-1.586 *** (0.308)	0.037 *** (0.004)	-0.025 *** (0.003)	
controls	Yes	Yes	Yes	
City FE	Yes	Yes	Yes	
Year FE	Yes	Yes	Yes	
Constant	55.711 *** (3.684)	0.140 *** (0.039)	-0.766 *** (0.174)	
N	4352	4352	4352	
R ²	0.938	0.610	0.877	

Table 3. Mechanism test.

Note: *** indicate significance levels of 1%. The values in parentheses are the robust standard error of regression coefficient.

5.2.2. Mechanistic Test for Industrial Structure

The mediating effect of industrial structure in the linkage between EID and urban PM2.5 emissions was further investigated, as reported in the fourth column of Table 3. The regression coefficient between EID and industrial structure is -0.025, which is significant at the 1% level, suggesting that EID notably reduces the secondary industry's share of national economies. Optimizing the industrial structure can be seen in the falling share of secondary industries and the rising share of tertiary-level industries in the national economy. The EID policy mitigates PM2.5 emissions by promoting industrial structure optimization, thus confirming Hypothesis 3.

The EID policy increases public awareness of environmental pollution, diminishes the positive perception of heavily polluting enterprises among urban residents, and discourages investment in and financing of such enterprises, ultimately reducing their economic profitability. In line with the capital's profit-maximizing motive, social resources in the market flow toward higher-profit enterprises, causing high energy-consuming and high-polluting industries to exit market competition, thereby decreasing the share of secondary industries in the national economy and driving industrial structure rationalization. As the industrial structure gradually optimizes, overall city pollution decreases, leading to improved control over PM2.5 emissions.

5.3. Heterogeneous Effects

The samples selected for this study encompass cities with diverse characteristics, reflecting significant variations in PM2.5 emissions. Given the variations in resource endowments and geographical locations across these cities, it is essential to examine the heterogeneity in the impact of EID policies on urban PM2.5 emissions.

5.3.1. Heterogeneity Test Based on Resource Endowment

Resource endowment refers to the diverse production factors available in a region. Significant differences in mineral resources across cities can influence their economic development models and, in turn, impact pollution emissions. The *National Sustainable Development Plan for Resource-Based Cities* (2013–2020), issued by the State Council, classifies cities as resource-based cities (RBCs) or non-resource-based cities (Non-RBCs). RBCs and Non-RBCs in China exhibit significant differences in industrial structures. Currently, heavy industries that predominantly consume fossil fuels are concentrated in RBCs, where pollution issues are more severe, potentially making these cities more responsive to environmental information disclosure policies. This study defines resource endowment as a

dummy variable, assigning a value of 1 to RBCs and 0 to Non-RBCs [80,81]. Among the 256 prefecture-level cities analyzed in this study, 107 are RBCs and 149 are Non-RBCs.

The regression results are displayed in columns 2 and 3 of Table 4. The regression coefficient between EID and PM2.5 emissions in resource-based cities is -3.105, passing the test at the 1% significance level, whereas in Non-RBCs, the coefficient is -0.328 and does not pass the significance test. This suggests that EID significantly reduces PM2.5 emissions in RBCs, but this effect is not verified in Non-RBCs. One potential explanation is that residents of resource-based cities are more aware of environmental pollution due to long-term exposure to haze, which directly affects their health. When the EID policy is implemented, pollution information is quickly disseminated and captures the attention of urban residents, creating strong public pressure that motivates local governments and enterprises to actively address environmental problems and reduce PM2.5 emissions. In contrast, residents of non-resource-based cities are less sensitive to publicly disclosed pollution information, which explains why EID has not yet shown a significant inhibitory effect on PM2.5 emissions in these cities.

Variable	Explained Variable: PM2.5				
vallable	RBCs	Non-RBCs	CACs	IACs	
did	-3.105 *** (0.508)	-0.328 (0.382)	-0.489 (0.396)	-1.928 *** (0.430)	
Controls	Yes	Yes	Yes	Yes	
City FE	Yes	Yes	Yes	Yes	
Year FE	Yes	Yes	Yes	Yes	
Constant	49.349 *** (5.735)	63.908 *** (6.075)	25.150 *** (6.765)	83.850 *** (8.463)	
Ν	1819	2533	1819	2533	
R ²	0.936	0.942	0.953	0.930	

Table 4. Heterogeneous effects results.

Note: *** indicate significance levels of 1%. The values in parentheses are the robust standard error of regression coefficient.

5.3.2. Heterogeneity Test Based on Urban Geographic Location

Considering China's vastness, significant differences in climate and economic development across regions may influence the impact of the EID policy on PM2.5 emissions. Thus, following existing research practices [82,83], the 256 sample cities were categorized into the following two groups based on geographic location: coastal area cities (CACs) and inland area cities (IACs). CACs were assigned a value of 1, and IACs were assigned a value of 0. The effects of EID on PM2.5 emissions across cities in different geographic locations were explored.

The results are displayed in the fourth and fifth columns of Table 4. The regression coefficients of EID on PM2.5 emissions in IACs is -1.928, passing the test at the 1% significance level, whereas in CACs, the coefficient is -0.489 and does not pass the significance test. This suggests that EID significantly reduces PM2.5 emissions in IACs, but this effect is not verified in CACs. IACs have lower levels of economic development and less established EID systems compared to coastal cities. Thus, the implementation of EID policy in inland cities can more effectively reduce PM2.5 emissions. In contrast, CACs are more economically developed, and environmental protection policies in these areas are more comprehensive, resulting in a less significant effect of environmental information disclosure.

5.4. Robustness Tests

To ensure the reliability of the conclusions, this section includes several robustness tests.

(1) Placebo test

To avoid spurious regression due to confounding policies and omitted variables, we employed an indirect placebo test. The basic premise is to construct virtual "treat" and

"post" variables to replace the actual "treat" and "post", generating a false estimate, β' . The resulting placebo test results after 500 repetitions are presented in Figure 3. These test results show that the estimator, β' , follows an approximate normal distribution with a mean coefficient of 0, which is distinctly different from the baseline regression coefficient. This outcome aligns with the expected results of the placebo test.



Figure 3. Placebo test results.

(2) Shortened sample period test

To identify the sensitivity of EID to temporal changes, we conducted a regression after shortening the sample period to 2006–2015. Meanwhile, other conditions remained unchanged. Column 2 in Table 5 indicates the regression results. We found that after shortening the sample period, the estimated coefficient is -0.896, significant at the 1% significance level. Compared to the baseline regression, the overall results did not change much, although the significance decreased. The results above show that changing the sample period affects the regression results, but our results are still noticeably negative. This further confirms the robustness of the results.

Variable	Explained Variable: PM2.5			
vallable	Shortened Period	Excluding Singular Values	Excluding Other Policy	
did	-0.896 *** (0.274)	-1.633 *** (0.274)	-1.659 *** (0.341)	
Controls	YES	YES	YES	
City FE	YES	YES	YES	
Year FE	YES	YES	YES	
Constant	60.003 *** (4.049)	54.797 *** (4.384)	70.638 *** (5.093)	
N	2,816	4352	3388	
R ²	0.959	0.939	0.940	

Table 5. Results of the robustness test.

Note: *** indicate significance levels 1%. The values in parentheses are the robust standard error of regression coefficient.

(3) Excluding singular values

Samples often contain individual data points that deviate from expectation, commonly referred to as outliers. If outliers are combined with normal data during econometric regression, the accuracy of the regression results may be compromised. To alleviate the effects of outliers, we shrunk the quartiles at 1% and 99% of the explained variable. The

regression coefficient of EID on PM2.5 is -1.633, which is negative and statistically notable at the 1% level, which verifies the reliability of the basic findings.

(4) Excluding other policies

Policy enforcement is inevitably affected by other policies, which may affect the accuracy of the results. To promote the Chinese economy's green transformation, the National Development and Reform Commission launched low-carbon city pilot projects in 2010 and expanded them further in 2012 and 2017. To eliminate the potential interference of these low-carbon city pilot policies with this study's findings, we categorized low-carbon pilot cities by implementation batch and re-ran the regression analysis. As reported in the fourth column of Table 5, the regression coefficient for EID remains negative, confirming the robustness of the baseline regression results.

(5) PSM-DID test

As most of the cities disclosing environmental information are key cities for environmental protection, the public list may have been selected from more polluted local cities. This nonrandom grouping makes it difficult to eliminate the interference of confounding variables and may lead to systematic bias. To overcome the systematic differences and reduce the fitting bias of the DID method, we used propensity score matching methods. First, we used logit regression to impute the propensity scores, with the control variables as the characteristic variables. Second, we used three methods, namely, 1:3 and 1:5 caliper nearest neighbor matching and kernel matching, to match the control group of cities affected by environmental information disclosure policies. Finally, we re-applied the DID method to estimate the net impact of EID on PM2.5. PSM can largely resolve the systematic bias of observable covariates, and DID can eliminate the effects of unobserved variables that are constant over time or change simultaneously over time. Combining these two methods can better detect the effects of external shocks. Table 6 displays the regression results after using PSM. Overall, we found that the net effect coefficients are all significantly negative, passing the 5% significance level tests. The above results do not alter the conclusion of the benchmark regression, so the robustness test is passed.

Variable	Explained Variable: PM2.5				
vallable	1:3	1:5	Kernel Matching		
did	-2.154 *** (0.678)	-1.893 *** (0.779)	-2.343 ** (0.926)		
Controls	Yes	Yes	Yes		
City FE	Yes	Yes	Yes		
Year FE	Yes	Yes	Yes		
Constant	57.006 *** (6.879)	58.403 *** (6.980)	58.552 *** (7.092)		
Ν	1893	1809	1734		
R ²	0.939	0.936	0.935		

Table 6. The results of PSM-DID tests.

Note: **, and ***, respectively, indicate significance levels of 5%, and 1%. The values in parentheses are the robust standard error of regression coefficient.

6. Conclusions and Policy Implications

The environmental information disclosure policy, as an institutional arrangement involving various stakeholders such as the government, corporations, and citizens, represents a new model of pollution management, wherein multiple parties coordinate to jointly reduce PM2.5 emissions. This study used China's 2007 Environmental Information Disclosure (Trial) policy as a quasi-natural experiment and applied a multiperiod DID model to empirically examine the net effect of EID on urban PM2.5 emissions, its underlying mechanisms, and the heterogeneity of its impact across cities with varying resources and geographic locations. The study's results indicate that, first, EID significantly reduces

urban PM2.5 emissions, and this conclusion remains robust after the parallel trend test and various robustness checks. Second, green innovation and industrial structure upgrades acted as key mediating channels through which EID reduces urban PM2.5 emissions. Third, regarding urban resource endowment heterogeneity, EID significantly reduces PM2.5 emissions in resource-based cities, whereas the effect is not significant in non-resource-based cities. From the perspective of geographic location, EID reduces PM2.5 emissions in inland area cities and is not as effective in coastal area cities.

The following policy suggestions emerge from the findings of the above research. First, it is essential to further enhance the environmental information disclosure system. This includes expanding the scope and content of EID, formulating unified disclosure standards and formats, facilitating convenient access to environmental information for all stakeholders, establishing environmental pollution reporting channels, and encouraging public participation in the identification and management of environmental pollution. Second, regarding the mediating channels of EID that affect urban PM2.5 emissions, the government should provide subsidies and tax breaks to encourage enterprises to increase R&D expenditures for green innovation; in addition, it should promote the closure and transformation of outdated production capacities, accelerate the development of green energy and energy-saving environmental protection industries, and optimize and upgrade the industrial structure through the gradual conversion of old and new kinetic energy. Finally, it is important to establish a differentiated assessment mechanism for the disclosure of urban environmental information and to implement policy incentives for resource-based and inland cities that demonstrate significant reductions in pollution, thereby further motivating these regions to reduce PM2.5 emissions.

This study examines the role of EID in pollution reductions at the city level, finding that EID effectively inhibits urban PM2.5 emissions. However, this study acknowledges certain research limitations and outlines directions for future research. On the one hand, this paper focuses on PM2.5 as a representative of pollution emission to explore the pollution reduction effect of EID. However, it should be pointed out that the daily production and life in the city will not only produce PM2.5, but also emit a large amount of carbon dioxide. The synergistic effect of pollution reduction and carbon reduction is an effective way to promote the green transformation of economic and social development. Therefore, the next step will be to explore the role of EID policy from the perspective of the synergy of pollution reduction and carbon reduction, which is an important role in further expanding the research field of EID and practicing urban green development. On the other hand, as the primary contributors to pollution emissions and the focal subjects of pollution management, it is crucial to investigate the impact of EID policies on enterprises and their pollution emissions to better understand the effectiveness of these policies. However, because of existing statistical standards, data related to enterprise-level pollution emissions, such as carbon emissions, sulfur dioxide, and PM, are currently unavailable. As micro-statistical data continue to improve, future studies on the impact of EID on corporate emissions at the enterprise level could significantly enrich this field.

Author Contributions: Conceptualization, T.W.; methodology, T.W. and Y.W.; software, Y.W.; data curation, Y.W.; writing—original draft preparation, Y.W.; writing—review and editing, T.W., Y.W. and W.X.; visualization, W.X.; supervision, T.W.; project administration, T.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The datasets generated during and/or analyzed during the current study are available from the corresponding author upon reasonable request.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Ministry of Ecology and Environment. The Ministry of Ecology and Environment Announces the Summary of National Ecological Environment Quality in 2019. Available online: https://www.mee.gov.cn/xxgk2018/xxgk/xxgk15/202005/t20200507_777895 .html (accessed on 1 June 2024).
- Ministry of Ecology and Environment of the People's Republic of China. 2019 China Ecological Environment Status Bulletin. Available online: https://www.mee.gov.cn/hjzl/sthjzk/zghjzkgb/202006/P020200602509464172096.pdf (accessed on 1 June 2024).
- 3. Tu, Z.; Shen, R. Can emissions trading scheme achieve the porter effect in China? *Econ. Res. J.* 2015, *50*, 160–173. (In Chinese)
- 4. Kathuria, V. Public disclosures: Using information to reduce pollution in developing countries. *Environ. Dev. Sustain.* 2009, 11, 955–970. [CrossRef]
- 5. Wang, K.; Yin, H.; Chen, Y. The effect of environmental regulation on air quality: A study of new ambient air quality standards in China. *J. Clean. Prod.* **2019**, *215*, 268–279. [CrossRef]
- Tao, W.; Song, M.; Weng, S.; Chen, X.; Cui, L. Assessing the impact of environmental regulation on ecological risk induced by PM2.5 pollution: Evidence from China. *J. Clean. Prod.* 2024, 451, 142029. [CrossRef]
- 7. Jephcote, C.; Chen, H. Environmental injustices of children's exposure to air pollution from road-transport within the model British multicultural city of Leicester: 2000–09. *Sci. Total Environ.* **2012**, *414*, 140–151. [CrossRef]
- Tao, Y.; Chai, Y.; Zhang, X.; Yang, J.; Kwan, M. Mobility-based environmental justice: Understanding housing disparity in real-time exposure to air pollution and momentary psycho-logical stress in Beijing, China. *Soc. Sci. Med.* 2021, 287, 114372. [CrossRef]
- 9. Yang, Z.; Xiong, Z.; Wang, L.; Xue, W. Can PM2.5 concentration reduced by China's environmental protection tax? *Sci. Total Environ.* **2024**, 937, 173499. [CrossRef]
- 10. Li, P.; Zhang, Z. The effects of new energy vehicle subsidies on air quality: Evidence from China. *Energy Econ.* **2023**, *120*, 106624. [CrossRef]
- 11. Zhao, Z.; Li, Y.; Su, X. Mountains block and seas move: The impact of geographical environment on the China's Carbon Emissions Trading Scheme in reducing urban PM2.5 concentrations. *Sustain. Cities Soc.* **2024**, *112*, 105630. [CrossRef]
- 12. Wang, R. Race to the top: Public oversight and local environmental information transparency in China. *Cities* **2024**, *148*, 104895. [CrossRef]
- 13. Feng, J.; Goodell, J.W.; Li, M.; Wang, Y. Environmental information transparency and green innovations. *J. Int. Financ. Mark. Inst. Money* **2023**, *86*, 101799. [CrossRef]
- 14. Brulle, R.; Turner, L.; Carmichael, J.; Jenkins, J. Measuring social movement organization populations: A comprehensive census of US environmental movement organizations. *Mobiliz. Int. Q.* **2007**, *12*, 255–270. [CrossRef]
- 15. Pan, X.W.; Fu, W.L. Environmental information disclosure and regional air quality: A quasi-natural experiment based on the PM 2.5 monitoring. *J. Financ. Econ.* 2022, *48*, 110–124. [CrossRef]
- 16. Zhou, Y.; Cao, J.; Feng, Y. Stock market reactions to pollution information disclosure: New evidence from the pollution blacklist program in China. *Sustainability* **2021**, *13*, 2262. [CrossRef]
- 17. Zhang, L.; Mol, A.P.; He, G.; Lu, Y. An implementation assessment of China's environmental information disclosure decree. *J. Environ. Sci.* 2010, 22, 1649–1656. [CrossRef]
- 18. Graham, M.; Miller, C. Disclosure of toxic releases in the United States. Environ. Sci. Policy Sustain. Dev. 2001, 43, 8–20. [CrossRef]
- Wang, T.; Peng, J.; Wu, L. Heterogeneous effects of environmental regulation on air pollution: Evidence from China's prefecturelevel cities. *Environ. Sci. Pollut. Res.* 2021, 28, 25782–25797. [CrossRef]
- 20. Williamson, C.R. Informal institutions rule: Institutional arrangements and eco-nomic performance. *Public Choice* 2009, 139, 371–387. [CrossRef]
- 21. Pargal, S.; Wheeler, D. Informal regulation of industrial pollution in developing countries: Evidence from Indonesia. *J. Polit. Econ.* **1996**, *104*, 1314–1327. [CrossRef]
- 22. Xu, B.; Xu, R. Assessing the role of environmental regulations in improving energy efficiency and reducing CO₂ emissions: Evidence from the logistics industry. *Environ. Impact Assess. Rev.* **2022**, *96*, 106831. [CrossRef]
- 23. Cao, Y.; Ren, W.; Yue, L. Environmental regulation and carbon emissions: New mechanisms in game theory. *Cities* **2024**, 149, 104945. [CrossRef]
- 24. Sinn, H.W. Public policies against global warming: A supply side approach. Int. Tax Public Financ. 2008, 15, 360–394. [CrossRef]
- 25. Zhang, K.; Zhang, Z.Y.; Liang, Q.M. An empirical analysis of the green paradox in China: From the perspective of fiscal decentralization. *Energy Policy* **2017**, *103*, 203–211. [CrossRef]
- Wang, H.; Zhang, R. Effects of environmental regulation on CO₂ emissions: An empirical analysis of 282 cities in China. Sustain. Prod. Consum. 2022, 29, 259–272. [CrossRef]
- 27. Tu, Z.; Hu, T.; Shen, R. Evaluating public participation impact on environmental protection and ecological efficiency in China: Evidence from PITI disclosure. *China Econ. Rev.* **2019**, *55*, 111–123. [CrossRef]
- 28. Zhang, H.; Xu, T.; Feng, C. Does public participation promote environmental efficiency? Evidence from a quasi-natural experiment of environmental information disclosure in China. *Energy Econ.* **2022**, *108*, 105871. [CrossRef]
- 29. Kou, P.; Han, Y.; Lin, B.; Li, T. How informal environmental regulations constrain carbon dioxide emissions under pollution control and carbon reduction: Evidence from China. *Environ. Res.* **2024**, 252, 118732. [CrossRef] [PubMed]

- 30. Shen, Q.; Pan, Y.; Feng, Y. Identifying and assessing the multiple effects of informal environmental regulation on carbon emissions in China. *Environ. Res.* **2023**, 237, 116931. [CrossRef]
- 31. Cui, H.; Cao, Y. How can informal environmental regulation improve urban air quality? Evidence from PITI publication in Chinese cities. *Urban Clim.* **2024**, *53*, 101813. [CrossRef]
- 32. Liu, Z.; Fang, C.; Sun, B.; Liao, X. Governance matters: Urban expansion, environmental regulation, and PM2.5 pollution. *Sci. Total Environ.* **2023**, *876*, 162788. [CrossRef]
- 33. Li, J.; Wu, T.; Liu, B.; Zhou, M. Can digital transformation enhance corporate ESG performance? The moderating role of dual environmental regulations. *Financ. Res. Lett.* **2024**, *62*, 105241. [CrossRef]
- 34. Lv, Y.; Wang, F.; Liu, G.; Ren, R. The impact of environmental court construction on the quality of corporate environmental information disclosure. *Int. Rev. Financ. Anal.* 2024, *95*, 103512. [CrossRef]
- 35. Zhang, Q.X.; Xiang, Z.Q. New media surveillance, environmental information uncertainty and corporate environmental information disclosure. *Int. Rev. Econ. Financ.* 2024, 95, 103477. [CrossRef]
- Chen, Y.; Masron, T.A.; Mai, W. Role of investor attention and executive green awareness on environmental information disclosure of Chinese high-tech listed companies. J. Environ. Manag. 2024, 365, 121552. [CrossRef] [PubMed]
- Zhang, J.; Zhang, L.; Zhang, M. Media pressure, internal control, and corporate environmental information disclosure. *Financ. Res. Lett.* 2024, 63, 105369. [CrossRef]
- 38. Zhang, C.; Zhou, B.; Wang, Q.; Jian, Y. The consequences of environmental big data information disclosure on hard-to-abate Chinese enterprises' green innovation. *J. Innov. Knowl.* **2024**, *9*, 100474. [CrossRef]
- 39. Wei, M.; Wang, Y.; Giamporcaro, S. The impact of ownership structure on environmental information disclosure: Evidence from China. *J. Environ. Manag.* 2024, 352, 120100. [CrossRef]
- 40. Yu, W.H.; Jin, X. Does environmental information disclosure promote the awakening of public environmental awareness? Insights from Baidu keyword analysis. *J. Clean. Prod.* 2022, 375, 134072. [CrossRef]
- Pan, A.; Qin, Y.; Li, H.; Zhang, W.; Shi, X. Can environmental information disclosure attract FDI? Evidence from PITI project. J. Clean. Prod. 2023, 403, 136861. [CrossRef]
- Wang, D.; Yang, W.; Geng, X.; Li, Q. Information disclosure, multifaceted collaborative governance, and carbon total factor productivity—An evaluation of the effects of the 'environmental information disclosure pilot' policy based on double machine learning. J. Environ. Manag. 2024, 366, 121817. [CrossRef]
- 43. Guo, X.; Xu, J. New ambient air quality standards, human capital flow, and economic growth: Evidence from an environmental information disclosure policy in China. *J. Clean. Prod.* **2024**, 434, 140168. [CrossRef]
- 44. Guo, C.; Jiang, Y.; Yu, F.; Wu, Y. Does environmental information disclosure promote or prohibit financialization of non-financial firms? Evidence from China. *Q. Rev. Econ. Financ.* **2023**, *92*, 200–214. [CrossRef]
- Feng, E.; Siu, Y.L.; Wong, C.W.Y.; Li, S.S.; Miao, X. Can environmental information disclosure spur corporate green innovation? Sci. Total Environ. 2024, 912, 169076. [CrossRef] [PubMed]
- Tan, X.; Liu, H. Corporate environmental performance under the pressure of government environmental information disclosure. *Financ. Res. Lett.* 2024, 69, 106152. [CrossRef]
- Feng, Y.; Chen, H.; Chen, Z.; Wang, Y.; Wei, W. Has environmental information disclosure eased the economic inhibition of air pollution? J. Clean. Prod. 2021, 284, 125412. [CrossRef]
- 48. Hu, D.; Huang, Y.; Zhong, C. Does environmental information disclosure affect the sustainable development of enterprises: The role of green innovation. *Sustainability* **2021**, *13*, 11064. [CrossRef]
- 49. Du, L.; Wang, F.; Tian, M. Environmental information disclosure and green energy efficiency: A spatial econometric analysis of 113 prefecture-level cities in China. *Front. Environ. Sci.* **2022**, *10*, 966580. [CrossRef]
- 50. Montmartin, B.; Herrera, M. Internal and external effects of R&D subsidies and fiscal incentives: Empirical evidence using spatial dynamic panel models. *Res. Policy* **2015**, *44*, 1065–1079. [CrossRef]
- Lu, Z.; Li, H. Does environmental information disclosure affect green innovation? *Econ. Anal. Policy* 2023, *80*, 47–59. [CrossRef]
 Liu, S.; Liu, C.; Yang, M. The effects of national environmental information disclosure program on the upgradation of regional
- industrial structure: Evidence from 286 prefecture-level cities in China. *Struct. Chang. Econ. Dyn.* **2021**, *58*, 552–561. [CrossRef] 53. Xuan, S.; Ge, W.; Yang, P.; Zhang, Y.F. Exploring digital finance, financial regulations and carbon emission nexus: New insight
- from resources efficiency, industrial structure and green innovation in China. *Resour. Policy* **2024**, *88*, 104452. [CrossRef]
- 54. Zhang, Q.; Li, J.; Kong, Q.; Huang, H. Spatial effects of green innovation and carbon emission reduction in China: Mediating role of infrastructure and informatization. *Sustain. Cities Soc.* **2024**, *106*, 105426. [CrossRef]
- 55. Wang, D.; Li, X.; Tian, S.; He, L.; Xu, Y.; Wang, X. Quantifying the dynamics between environmental information disclosure and firms' financial performance using functional data analysis. *Sustain. Prod. Consum.* **2021**, *28*, 192–205. [CrossRef]
- Dhaliwal, D.S.; Li, O.Z.; Tsang, A.; Yang, Y.G. Voluntary nonfinancial disclosure and the cost of equity capital: The initiation of corporate social responsibility reporting. *Acc. Rev.* 2011, *86*, 59–100. [CrossRef]
- Flammer, C. Corporate social responsibility and shareholder reaction: The environmental awareness of investors. *Acad. Manag. J.* 2013, 56, 758–781. [CrossRef]
- Li, W. Self-motivated versus forced disclosure of environmental information in China: A comparative case study of the pilot disclosure programmes. *China Q.* 2011, 206, 331–351. [CrossRef]

- 59. Driessen, P.H.; Hillebrand, B.; Kok, R.A.; Verhallen, T.M. Green new product development: The pivotal role of product greenness. *IEEE Trans. Eng. Manag.* 2013, *60*, 315–326. [CrossRef]
- 60. Wang, D.; Liu, X.; Yang, X.; Zhang, Z.; Wen, X.; Zhao, Y. China's energy transition policy expectation and its CO₂ emission reduction effect assessment. *Front. Energy Res.* **2021**, *8*, 627096. [CrossRef]
- 61. Chen, J.; Gao, M.; Mangla, S.K.; Song, M.; Wen, J. Effects of technological changes on China's carbon emissions. *Technol. Forecast. Soc. Chang.* **2020**, *153*, 119938. [CrossRef]
- 62. Chen, Y.J.; Li, P.; Lu, Y. Career concerns and multitasking local bureaucrats: Evidence of a target-based performance evaluation system in China. J. Dev. Econ. 2018, 133, 84–101. [CrossRef]
- 63. Keller, W.; Levinson, A. Pollution abatement costs and foreign direct investment inflows to US states. *Rev. Econ. Stat.* 2002, *84*, 691–703. [CrossRef]
- 64. Wan, K.; Yu, X. Optimal governance radius of environmental information disclosure policy: Evidence from China. *Econ. Anal. Policy* **2024**, *83*, 618–630. [CrossRef]
- 65. Tang, P.; Jiang, Q.; Wang, C. Beyond environmental actions: How environmental regulations stimulate strategic-political CSR engagement in China? *Energy Econ.* **2024**, *129*, 107171. [CrossRef]
- Engel-Cox, J.A.; Young, G.S.; Hoff, R.M. Application of satellite remote-sensing data for source analysis of fine particulate matter transport events. J. Air Waste Manag. Assoc. 2012, 55, 1389–1397. [CrossRef] [PubMed]
- 67. Wang, S.; Xu, L.; Ge, S.; Jiao, J.; Pan, B.; Shu, Y. Driving force heterogeneity of urban PM2. 5 pollution: Evidence from the Yangtze River Delta, China. *Ecol. Indic.* 2020, 113, 106210. [CrossRef]
- 68. Chen, J.; Gao, M.; Cheng, S.; Hou, W.; Song, M.; Liu, X.; Liu, Y. Global 1 km × 1 km gridded revised real gross domestic product and electricity consumption during 1992–2019 based on calibrated nighttime light data. *Sci. Data* **2022**, *9*, 202. [CrossRef]
- 69. Wang, S.; Wang, H.; Wang, J.; Yang, F. Does environmental information disclosure contribute to improve firm financial performance? An examination of the underlying mechanism. *Sci. Total Environ.* **2020**, *714*, 136855. [CrossRef]
- Yu, Z.; Yan, T.; Liu, X.; Bao, A.Z. Urban land expansion, fiscal decentralization and haze pollution: Evidence from 281 prefecturelevel cities in China. J. Environ. Manag. 2022, 323, 116198. [CrossRef]
- Cheng, Z.; Li, L.; Liu, J. The impact of foreign direct investment on urban PM2.5 pollution in China. J. Environ. Manag. 2020, 265, 110532. [CrossRef]
- 72. Wang, F.; He, J.; Niu, Y. Role of foreign direct investment and fiscal decentralization on urban haze pollution in China. *J. Environ. Manag.* **2022**, *305*, 114287. [CrossRef]
- Uddin, M.; Siddik, A.B.; Zhao, Y.H.; Naeem, M.A. Fintech and environmental efficiency: The dual role of foreign direct investment in G20 nations. J. Environ. Manag. 2024, 360, 121211. [CrossRef] [PubMed]
- Luo, X.; Sun, K.; Li, L.; Wu, S.M.; Yan, D.; Fu, X.S.; Luo, H. Impacts of urbanization process on PM2.5 pollution in "2 + 26" cities. J. Clean. Prod. 2021, 284, 124761. [CrossRef]
- 75. Dong, Q.; Lin, Y.; Huang, J.; Chen, Z.F. Has urbanization accelerated PM2. 5 emissions? An empirical analysis with cross-country data. *China Econ. Rev.* 2020, *59*, 101381. [CrossRef]
- Fu, Y.; Zhang, Y. Chinese decentralization and fiscal expenditure structure bias: The cost of competition for growth. *Manag. World* 2007, 3, 4–12. [CrossRef]
- 77. Li, Q.; Wang, Y.; Chen, W.; Li, M.; Fang, X. Does improvement of industrial land use efficiency reduce PM2.5 pollution? Evidence from a spatiotemporal analysis of China. *Ecol. Indic.* **2021**, *132*, 108333. [CrossRef]
- 78. Su, W.; Xie, C. High-speed rail, technological improvement, and PM2. 5: Evidence from China. *Econ. Anal. Policy* 2023, 80, 1349–1362. [CrossRef]
- 79. Jiang, N.; Jiang, W.; Zhang, J.; Chen, H. Can national urban agglomeration construction reduce PM2.5 pollution? Evidence from a quasi-natural experiment in China. *Urban Clim.* **2022**, *46*, 101302. [CrossRef]
- 80. Liang, C.L.; Chen, X.L.; Di, Q.B. Path to pollution and carbon reduction synergy from the perspective of the digital economy: Fresh evidence from 292 prefecture-level cities in China. *Environ. Res.* **2024**, 252, 119050. [CrossRef]
- Lin, B.; Xu, C. Enhancing energy-environmental performance through industrial intelligence: Insights from Chinese prefecturallevel cities. *Appl. Energy* 2024, 365, 123245. [CrossRef]
- 82. Jin, P.Z.; Wang, S.Y.; Yin, D.S.; Zhang, H. Environmental institutional supply that shapes a green economy: Evidence from Chinese cities. *Technol. Forecast. Soc. Chang.* 2023, 187, 122214. [CrossRef]
- Liu, W.; Gao, Y.; Tang, O.; Cheng, Y. Comprehensive performance analysis of deep integration and innovative development of logistics and manufacturing industries: A comparison analysis between coastal and inland regions in China. *Ocean Coast. Manag.* 2024, 257, 107332. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.